Title: Sexual assault in Lagos, Nigeria: a five year retrospective review.

Authors:

Fatimat M Akinlusi (fatimatakinlusi@yahoo.co.uk)
Kabiru A Rabiu (derabs@yahoo.co.uk)
Tawakalit A Olawepo (botawakkalt@yahoo.com)
Adeniyi A Adewunmi (niyi_55@yahoo.com)
Tawakalit A Ottun (taottun@yahoo.com)
Oluwarotimi I Akinola (iretiakinola@hotmail.com)

Version: 3 Date: 11 September 2014

Author’s response to reviews:

Dear Editor,

Re: Sexual assault in Lagos, Nigeria: a five year retrospective review

Thank you for facilitating the re-review of the above named manuscript.

The corrections suggested by Reviewer 2 have been implemented.

Please find below a point by point response to his concerns.

Minor essential Revision:

In discussion section, "we" and "our" have been deleted and the sentences affected modified appropriately.

Case definition and case identification have been embedded under "study population"

Also, "Limitation" is now part of "Discussion"

Other language corrections have been effected.

Yours Sincerely,

Fatimat M Akinlusi

Akinlusi Fatimat M.